Author’s response to reviews

Title: Factors involved in nurses’ responses to burnout: a grounded theory study

Authors:

Forough Rafii (foroughrafii@yahoo.com)
Fatemeh Oskouie (nmsi@iums.ac.ir)
Mansoure Nikravesh (manyunik@yahoo.com)

Version: 3 Date: 8 November 2004

Author’s response to reviews:

Dear Editorial board
Hereto attached is the revised manuscript with the title: "Factors intervene in nurses' responses to
burnout in burn centers of Tehran: a grounded theory study" by Forough Rafii, Fatemeh Oskouie, and
Mansoure Nikravesh.
The manuscript is new and is not under consideration for publication elsewhere. Copyright to the manuscript
will be assigned to the journal, when published.
The following changes have made to the manuscript:
1. The title was changed as the reviewers' comments and the first page was considered for this purpose.
The new title is: "factors involved in nurses' responses to burnout: a grounded theory study".
2. Recommended changes were made to the abstract and it was started in the second page.
3. The first paragraph of the background was removed.
4. Grounded theory was defined in the methods section.
5. Tables were placed as near as possible to their first mention in the manuscript and their titles were placed
above them. The format of table 5 was changed and individual cells were used instead of a diagonal slash.
6. It was stated in the data analysis section that quotes from participants are exact translations of their
words.
7. It was declared that the authors have no competing interests. This statement was placed between the
conclusions and authors' contributions.
8. A copy of the manuscript was sent for one of our colleagues in US and she edited the copy completely.
I hope that we could have made the needed changes in light of the reviewers' comments and instructions to
fulfil the scope of the BMC nursing.
I'm looking forward for your response.
Sincerely yours

Forough Rafii/
On behalf of all authors

Correspondence address:
Forough Rafii, Doctoral student's
Department, Faculty of Nursing and Midwifery,
Rashid Yasami St., Vali-Asr Ave., Tehran, 19395-4798 Iran
Tel: +98 21 8773073
Fax: +98 21 8793805
E-mail: foroughrafii@yahoo.com